Author's response to reviews

Title: Height, weight and BMI percentiles and nutritional status relative to the international growth references among Pakistani school-aged children

Authors:

Muhammad Umair Mushtaq (mushtaqmu@gmail.com)
Sibgha Gull (sibgha_gul@ymail.com)
Komal Mushtaq (komalmushtaq11@gmail.com)
Hussain Muhammad Abdullah (abdullahh_aimc@yahoo.com)
Usman Khurshid (usmankhurshid_aimc@yahoo.com)
Ubeera Shahid (mushtaqmu@gmail.com)
Mushtaq Ahmad Shad (mushtaq_shad@yahoo.com)
Javed Akram (jakramaimc@gmail.com)

Version: 3 Date: 20 January 2012

Author's response to reviews:

Previously the results in the manuscript were presented by gender-stratification. Data for total sample is added, and the paper is revised accordingly to make it more comprehensive (there are no deletions, this revised version contain additions).

Thank you all for your time and efforts to improve the manuscript.